Author's response to reviews

Title: The use of opioids at the end of life: a survey of experienced barriers in pain management and knowledge level of physicians

Authors:

Mette L Rurup (m.rurup@vumc.nl)
Christiaan A Rhodius (crhodius@gmail.com)
Sander D Borgsteede (s.borgsteede@gmail.com)
Manon SA Boddaert (mboddaert@amsteling.nl)
Astrid GM Keijser (astrid.keijser@gmail.com)
H Roeline W Pasman (hrw.pasman@vumc.nl)
Bregje D Onwuteaka-Philipsen (b.philipsen@vumc.nl)

Version: 6 Date: 11 November 2010

Author's response to reviews:

Dear editor,
I have made the asked revisions:
- used the appropriate format for citing appendix 1
- all references are cited in numerical order in the text, including reference 61.
- keywords and running head are removed from the abstract
- reference 25 and 27 are completed.
Sincerely,
Bregje Onwuteaka-Philipsen